



Pseudolites to Support Location Services in Smart Cities: Review and Prospects

Tong Liu ^{1,2,3}, Jian Liu ^{2,4}, Jing Wang ⁵, Heng Zhang ⁶, Bing Zhang ^{2,3}, Yongchao Ma ², Mengfei Sun ², Zhiping Lv ² and Guochang Xu ^{1,2,7,*}

- Key Laboratory of Urban Land Resources Monitoring and Simulation, Ministry of Natural Resources, Shenzhen 518000, China; 19b958024@stu.hit.edu.cn
- ² Institute of Space Science and Applied Technology, Harbin Institute of Technology (Shenzhen), Shenzhen 518000, China; jianliu@stu.hit.edu.cn (J.L.); zhangbing@stu.hit.edu.cn (B.Z.);
- ma_yongchao@stu.hit.edu.cn (Y.M.); 20b958036@stu.hit.edu.cn (M.S.); lvzhiping@hit.edu.cn (Z.L.)
 ³ Department of Land Surveying and Geo-Informatics, The Hong Kong Polytechnic University, Hong Kong 999077, China
- ⁴ Department of Aeronautical and Aviation Engineering, The Hong Kong Polytechnic University, Hong Kong 999077, China
- ⁵ Wuhan Mengxin Technology Co., Ltd., Wuhan 430073, China; jing.wang@wh-mx.com
- ⁶ State Key Laboratory of Satellite Navigation System and Equipment Technology, The 54th Research Institute of China Electronics Technology Group Corporation, Shijiazhuang 050081, China; zhhcetc@163.com
- ⁷ Shenzhen Institute of Advanced Technology, Chinese Academy of Sciences, Shenzhen 518000, China
- Correspondence: xuguochang@hit.edu.cn; Tel.: +86-183-8910-1556

Abstract: The location service is an important part of the smart city. A unified location service for outdoor and indoor/overground and underground activity will assist the construction of smart cities. However, with different coordinate systems and data formats, it is difficult to unify various positioning technologies on the same basis. Global navigation satellite system (GNSS)-based positioning is the only way to provide absolute location under the Earth-centered, Earth-fixed coordinate system (ECEF). Increasing indoor and underground human activity places significant demand on location-based services but no GNSS signals are available there. Fortunately, a type of satellite that is indoors, known as pseudolite, can transmit GNSS-like ranging signals. Users can obtain their position by receiving ranging signals and their resection without adding or switching other sensors when they go from outdoors to indoors. To complete the outreach of the GNSS indoors and underground to support the smart city, how to adapt the pseudolite design and unify coordinate frames for linking to the GNSS remain to be determined. In this regard, we provide an overview of the history of the research and application of pseudolite-based location services in smart cities.

Keywords: pseudolite; indoor positioning; spatial datum; seamless navigation; GNSS; PNT; Internet of Things; integrated navigation

1. Introduction

Smart cities place a significant demand on accurate location services. End-users of smart cities—typically via smartphones, vehicles, wearables, etc., with respect to the Internet of Things (IoT)—require precise location and time information [1]. For positioning, a reference datum is needed first. Positioning techniques such as ultra-wide band (UWB) [2], received signal strength indication (RSSI) [3], ZigBee [4], etc., could provide precise positioning, but only in a local coordinate system. Different from other indoor positioning methods, the GNSS receiver is the only sensor that can provide absolute positioning in the geocentric coordinate system. Thus, GNSS-based spatial data [5] could be regarded as the smart city's infrastructure for common positioning and navigation services.



Citation: Liu, T.; Liu, J.; Wang, J.; Zhang, H.; Zhang, B.; Ma, Y.; Sun, M.; Lv, Z.; Xu, G. Pseudolites to Support Location Services in Smart Cities: Review and Prospects. *Smart Cities* 2023, *6*, 2081–2105. https://doi.org/ 10.3390/smartcities6040096

Academic Editor: Pierluigi Siano

Received: 18 July 2023 Revised: 8 August 2023 Accepted: 15 August 2023 Published: 18 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Unfortunately, the GNSS can be used only in an open environment on the ground surface since the L-band signal of the GNSS cannot be transmitted directly to the subsurface. Thus, it is hard to rely only on the GNSS to complete underground data construction. Indoor positioning technologies such as UWB [6] have system differences from the GNSS and require additional receiving sensors for users [7]. Moreover, they could only provide customized relative locations in a free local coordinate system, rather than the absolute location in the Earth's coordinate system. In this case, the current indoor positioning techniques make it difficult to cooperate effectively with GNSS. Now, the indoor and underground location services based on GNSS spatial data are still lacking [8,9].

Pseudolites can emit GNSS-like signals, thus enabling continuous positioning both indoors and outdoors. If the pseudolite uses a standard-format GNSS frequency and signal, then the GNSS receiver, without any modification, can process the pseudolite signal [10]. In comparison with other methods, the pseudolite is able to provide a datum associated with the GNSS for indoor and underground positioning and seamless navigation.

In recent years, researchers have studied pseudolites from many different perspectives, including system commissioning, signal transmission, signal reception, error processing, and pseudolite-based integrated navigation. Pseudolites are becoming closer to large-scale applications, especially with the introduction of hardware solutions for signal design and reception in the field of communications, where the problems of near–far effects and clock synchronization are largely solved.

However, how to achieve the location services for smart cities under the GNSS via pseudolite has rarely been discussed. Moreover, as a technology with great potential applications, few authors have reviewed and analyzed the technical progress, existing problems, and prospects of pseudolites. The last pseudolite review article was published 20 years ago [11]. In the last 20 years, many advances in pseudolite technology have occurred and higher demand has been placed on the technology.

Facing the unification of spatiotemporal reference and seamless navigation, higher requirements for the pseudolite system's construction has become apparent. This is related to two issues: on the one hand, the unification of orbit is indispensable; current pseudolites are only used for indoor positioning, while seamless navigation requires that the orbits of pseudolites are integrated into the coordinate system of the GNSS, such as the "World Geodetic System 1984" (WGS84); on the other hand, there is the pseudolite constellation; current research is mostly tested in open indoor environments or with a well-indoor pseudolite constellation; however, there are random and multiple masks in real indoor environments such as in shopping malls which lead to poor positioning accuracy. Facing the problem of obscured signals, how to better optimize the constellation structure of pseudolites and combine multiple sensors is the key issue in making high positioning accuracy obtainable in a larger indoor area [6,11].

To further promote the application of pseudolite technology in smart cities and to serve the construction of the positioning, navigation, and timing (PNT) system, it is necessary to sort out and present the current status of pseudolite technology and its development trend. In this review, Section 2 provides a brief introduction to the development and applications of pseudolites; Section 3 reviews the status architecture and challenges on the pseudolite system side; Section 4 presents the status of the main academic and technical inspections on the receiver side; Section 5 provides a discussion of the current issues and the prospects; Section 6 describes the possible role of pseudolites in smart cities; the final section provides the conclusion.

2. Brief Development History and Existing Applications

As early as the construction of the GPS in the United States, research on pseudolite problems has been carried out. A schematic diagram of the workflow of a single pseudolite with both receiving and transmitting functions can be seen in Figure 1. The earliest pseudolite started from simulating the GPS L1 signal [12]. After that, the theory of pseudolites for positioning and navigation was discussed [13–15]. To verify the feasibility of GPS theory, a pseudolite test site in the United States named Yuma was built [16]. At that time, the above studies showed that pseudolites were used more as a technology to test GPS rather than a stand-alone positioning technology. After a period of development, by 2002, the progress of GPS-based pseudolites and the technical problems existing at that time were reviewed [11].



Figure 1. Schematic diagram of the workflow of a single pseudolite with both receiving and transmitting functions.

In addition to those who built the GPS, other GNSS communities have also worked on pseudolites. A Galileo satellite navigation system (Galileo) experimental test environment named GATE was built in Germany, and positioning tests were carried out by transmitting Galileo signals from six ground-based pseudolites [17]. A pseudolite signal-transmitting system based on GPS L1 signals produced by a Finnish company was tested. Although the positioning error can sometimes reach tens of meters, seamless navigation was initially achieved [18]. Pseudolites were used as an augmentation tool during the development of the European Geostationary Navigation Overlay Service (EGNOS) [19,20]. With the development of BeiDou, the airborne BeiDou-capable pseudolite system was developed [21]. Pseudolite systems combining BeiDou [22] and dual BeiDou/GPS systems [23] were developed. Along with the development of the GNSS, pseudolites have been developed for more than 40 years.

Research has shown that pseudolite positioning is a feasible technology and is applied in many fields. The application scenarios of pseudolites include location services for the public and exclusive applications for industries, as shown in Figure 2. To the best of our knowledge, the application scenarios of pseudolite positioning around location services for smart cities, in addition to common indoor positioning, include positioning of aircraft, positioning of trains or subways, underwater positioning, deformation monitoring, and deep space exploration.



Figure 2. Schematic of pseudolite applications in smart cities.

2.1. For Aircraft Positioning

Currently, the hotspot of GNSS research in civil aviation is the theory, methods, and experiments of augmentation systems, including satellite-based augmentation systems (SBASs), ground-based augmentation systems (GBASs), and aircraft-based augmentation systems (ABASs). Since the pseudolites could not achieve the equivalent performance as GNSS augmentation systems required by civil aviation, in comparison with the augmentation systems, pseudolite-based civil aviation applications have received less attention in recent years but were studied earlier.

For civil aviation, in addition to being a tool for GPS testing, pseudolites are used for GPS augmentation, for example, to assist in aircraft approaches [24–30]. An integrated GPS, GLONASS, and pseudolite navigation approach was proposed for Category III approaches and landings [31]. A pseudolite-based augmentation system (PBAS) for civil aviation navigation was introduced. A test in Korea showed that the PBAS could improve positioning accuracy by approximately 50% [32,33].

Pseudolite applications for unmanned aerial vehicles (UAVs) were investigated. When there is a good geometry of the pseudolite placement, the performance is comparable to GPS performance [34]. Aeronautical applications of integrated navigation using pseudolites have been explored, and the results show that pseudolites help improve the vertical accuracy of aircraft [35,36]. In the aerodrome zone, the positioning geometry factor of pseudolites with GLONASS orbit simulation is discussed. Results show that the number and placement of pseudolites are the keys to improving the accuracy [37]. Moreover, pseudolites are used as a ranging method, and the results show that the clock synchronization performance is at least better than that of other distance measurement equipment (DME), though the data reception quality is slightly worse [38]. For outdoor civil aviation or UAV positioning, there is a trade-off between the improved positioning accuracy from the pseudolite and the ranging error introduced by the associated clock deviation [39]. Airborne BeiDou pseudolites have been preliminarily discussed [21,40]. At present, the exploration of ground-based BeiDou-pseudolite-based civil aviation is still almost non-existent.

2.2. For Train Positioning

The optimization of the constellation design of pseudolite positioning at railway stations was investigated [41], and an integrated train positioning system based on an integrated GNSS/pseudolite model was proposed. The performance of optimized pseudolite constellations and their impact on GNSS/pseudolite-based seamless positioning for trains was demonstrated through a case study involving a reference pseudolite constellation strategy [42]. Both studies show the importance of the placement of pseudolites at train stations or trackside. In addition to positioning on the ground, geostationary pseudolites and Doppler algorithms could be used underground for the subway [43]. Although there are rare train positioning solutions using pseudolites, these studies have demonstrated the potential of pseudolite applications in train positioning.

2.3. For Underwater Positioning

As early as 1991, the idea of pseudolite application for underwater positioning was proposed [44]. Due to severe signal attenuation in water, the attenuation model of signal transmission needs to be addressed [45]. Studies show that it is more appropriate to use acoustic and other means of localization underwater [46,47].

Although pseudolites cannot be used directly for underwater positioning, pseudolitelike devices on the surface can transmit time and space references underwater, which is of great importance for underwater mapping and navigation. As a position reference rather than a positioning method, pseudolites can be used as buoys for underwater localization [48]. Moreover, pseudolites can be used for the mutual correction of positions between buoys on the water to determine a dynamic surface positioning reference [49].

2.4. For Deformation Monitoring

Usually, for deformation monitoring of dams, reservoirs, slopes, and bridges, the GNSS is sufficient. There have been many practical cases of the GNSS being used for landslide monitoring [50–52]. However, pseudolites can still be involved due to their advantages in canyons or tree-shaded environments; for example, they can be used for ground settlement monitoring in urban canyons to supplement and validate interferometric synthetic aperture radar (InSAR) monitoring results [53,54]. A combination of GPS and pseudolites is proposed to monitor deformation in urban canyons areas. The addition of pseudolites improves the accuracy of deformation monitoring compared to GPS use alone [55]. The structural deformation of bridges was monitored using the GPS and pseudolites; when the geometry is well and multipath effects are mitigated, a millimeter-level deformation monitoring accuracy is achievable [56].

Pseudolites can also be used in conjunction with a variety of other sensor combinations for deformation monitoring [57]. For deformation monitoring using pseudolites, it is more important to determine the relative displacement rather than the absolute position, which leads to the possibility of using a different technical route than that of GNSS positioning. For example, since the pseudolite is stationary, it is possible to monitor the deformation by ranging the distance from the pseudolite transmitter to the receiver located at the displacement.

2.5. For Deep Space Positioning

In the lunar exploration project, the accuracy geometry factor could be reduced by setting pseudolites when conducting positioning on the lunar surface using the GNSS [58]. A local-area GPS pseudolite-based navigation system can be used for Mars exploration [59]. If pseudolites are deployed over a wide area on the Moon or Mars, they can assist in the orbit determination of the probe and navigation for landing [60,61].

2.6. Cases of Practical Application

Smart cities rely on outdoor GNSS positioning. Location services in smart cities are limited in GNSS-denied environments. However, several cases have shown that pseudolites can be successfully applied in GNSS denial environments.

2.6.1. The Locata System

Locata is one of the most established pseudolite systems. It was initially designed to address the multipath effect under urban canyons and was later applied indoors [62,63]. The system features a nanosecond time synchronization program [64]. In practical engineering, the Locata and GPS time synchronization performed well [65]. In its signal coverage area, Locata's positioning accuracy is much better than that of the GPS [66], especially in the up direction [67]. In particular, Locata technology can provide sub-centimeter indoor positioning in the horizontal direction by mitigating cycle slips and multipath effects [68].

In recent years, with the enhancement of processing clock differences [69,70], cycle slips [71], tropospheric delays [72], and multipath [62,73], Locata has gradually been increasingly used in more applications, such as open-cut mining [74,75], deformation monitoring [67,76], flight testing [77–81], and integrated navigation [80,81].

2.6.2. The Beidou/GPS Dual-Mode System

The 54th Institute of China Electronics Technology Group Corporation (CETC), alongside the Wuhan Mengxin Technology Co., Ltd., have developed dual-mode pseudolites and their signal receiver that can accept BeiDou and GPS signals, as shown in Figure 3. This system mainly consists of receiving antenna, forwarding antenna, time synchronization ring, and pseudolite host. A detailed presentation is available [23]. The main technical aspects include the signal design [82] of BeiDou and GPS, the processing of near–far effects [83], closed-loop clock synchronization, and integrated positioning. The pseudolite device was used for indoor positioning of the "Snow Ruyi" venue for the Beijing Winter Olympics.



Figure 3. Pseudolite Receiver: module and integrated unit.

Recently, a multi-PNT source-based positioning system [84] has been designed for GNSS signal denial environments. Its infrastructure includes mobile communication base stations, Wi-Fi networks, UWB transmitters, radio frequency identification (RFID) transmitters, and pseudolites. Its elastic PNT terminal includes a GNSS receiver, inertial unit, Wi-Fi module, 5G module, Bluetooth module, and lidar. The software embedded in the device functions for data capture, fusion, and localization. The system has been tested with sub-meter accuracy [85] and has been successfully used in the Yan Chong Expressway, which contains several tunnels.

2.7. An Underlying Positioning Technology for the Internet of Things

Alongside the outdoor applications mentioned above, in recent years, pseudolites have been considered popular candidates for indoor positioning and seamless indoor-outdoor navigation. Hot research topics include technical issues in simulating or transmitting signals, and signal- or data-oriented processing methods to improve positioning accuracy. The Internet of Things (IoT) requires location services as its underlying technology support [86]. The positioning error sources of the IoT have been analyzed [87]. A prototype of location services for the IoT has been described [88], which essentially enables seamless indoor and outdoor navigation without the addition of other indoor positioning devices. In the IoT, pseudolites can be used as a key technology for positioning, which can provide high accuracy and GNSS connections to the outdoors. The establishment of a spatial datum based on pseudolites can realize the unification of ground–underground and indoor–outdoor spatial data and provide GNSS-like accurate location services for underground/indoor operations in megacities. In this vision, dynamic and static positioning of indoor pedestrians, cell phones, bracelets, pendants, and other belongings can be achieved, for example, in a scheme for baby positioning [89].

Location and telecommunication are the two key elements of the IoT [90]. As an essential part of IoT applications, pseudolites need to be integrated with communication solutions such as 5G [91,92] in order to further expand their capabilities. One existing success case is the construction of a low-cost regional navigation system [93,94] based on decommissioned communications satellites.

3. Architecture and Technical Issues on the System Side

The current state of research on pseudolites is divided into technical problems that could be solved on the hardware system side and the main academic/algorithmic problems to be solved at the receiver side. The issues related to pseudolite signal transmission, clock synchronization problems, and orbit problems that can be optimized from the system side are first discussed, after which the received data and their processing refinement can be addressed on the receiver side.

3.1. Signal Intensity and Near–Far Effect

Receiving a sufficiently strong signal is a prerequisite for a receiver to locate; therefore, the receiver should be able to capture weak or reflected signals at a distance [95,96]. A time-hopping (TH) pulse position detection method [97] solves the problem of low signal detection success rate in a low signal-to-noise environment. To mitigate the effects of signal degradation, a robust and general time-hopping direct sequence spread spectrum (TH-DSSS) signal acquisition method was proposed [98]. When setting up the new signal method, the number of operations should be considered at the same time to avoid low timeliness due to computational overload [99,100].

The attenuation of electromagnetic waves in the air is large. Pseudolite signals near the receiver can create power suppression of distant signals, leading to the near–far effect, as demonstrated in Figure 4. Resolving the near–far effect is the primary requirement for successful pseudolite positioning.

Traditionally, the near–far effect may be mitigated by the following [11,101]: (1) transmitting a pulse signal with a fixed period in the GPS band but with frequency offset; this scheme is feasible because by adjusting the carrier signal frequency, the pseudolite can be made to operate in a variable frequency band determined by the signal frequency [102]; (2) transmitting pseudolite signals at the frequency offset from the GPS, but within the same frequency band, or using a longer coding sequence than the GPS. In addition, a successive interference cancellation method [95] allows GPS receivers to receive pseudolite signals at 30 dB to 40 dB. New programs have been successively proposed over recent years. A general pulsing scheme based on random permutations was proposed to mitigate this effect [103]; signal orthogonality can be exploited to counteract the pseudolite signal near–far effect [104].



Figure 4. From the ground, different user locations and the corresponding received signal propagation paths: User B is in the near zone of pseudolite C, whose signal suppresses the distant signal of pseudolite D; User A is in the middle region of pseudolite C and can receive equal pseudolite C and distant signals pseudolite D; and in the far field of pseudolite C, the signal of pseudolite C is suppressed.

3.2. Clock Synchronization

Time synchronization is a necessary issue to be stressed for seamless navigation [105,106] since unsynchronized clocks can introduce significant positioning errors. For specific positioning algorithms and operational requirements, a targeted analysis is required to determine a reasonable solution for introducing pseudolites.

The impact of clock offset on both single-point positioning [107] and differential positioning [108] is significant. Clock discrepancies from ground-based transmitters can lead to asynchronous timing problems [109] and hinder attempts to fix integer ambiguity [110]. Unfortunately, pseudolites cannot be time-synchronized by expensive satellite-based clocks such as the GNSS, and only low-cost solutions can be adopted.

Wireless synchronization methods are favored for their flexibility in system deployment [111,112]. A tree topology for time information flow can be constructed based on the master node; thus, a pseudolite can receive signals broadcast by multiple pseudolites. For example, with the Locata system, a master transceiver synchronizes its slave transceivers over a wireless network [69]. Based on this approach, bidirectional time synchronization between multiple pseudolites over a network cable or wirelessly can be achieved [107,113].

Considering the problem of time synchronization in pseudolites, indoor precise point positioning (iPPP) [114] and iPPP with real-time kinematic (iPPP-rtk) [110] systems were proposed to address time deviation. They can reliably achieve accurate indoor positioning at the centimeter level. Time delay fluctuations can disrupt the integer properties of the carrier phase. A fractional cycle deviation correction method [107] was proposed for the case where the ambiguity of observations in pseudolites is usually estimated as a floating-point value. This method is based on bidirectional time synchronization and corrects its own bias by pseudolite network estimation.

The clock can remain synchronized in the presence of link outages or pseudolite failures. A mesh topology clock synchronization (MTCS) technique based on a mesh topology using all received signals was proposed [115]. MTCS constructs a mesh topology for time information flow, derives the coupling relationship from the clock. In addition, the temporal deviation can be estimated by putting it into the process of position decomposition in the case of clock link interruption [116]. From a hardware point of view, in addition to the use of networks, fiber optics is also a solution that can be used for clock synchronization [117–119].

Current research shows that clock synchronization only pseudolites only can be achieved through hardware and software treatments. However, the time synchronization between pseudolites and real satellites needs to the focus if the time systems defined by them are expected to be the same. In the future, the issue could be further addressed at both the system and user levels. On the hardware level, optical fibers can be used to connect the ground network of pseudolites to the BeiDou ground system [118]; on the common user level, a clock transfer technique is available [120]; for example, the Coordinated Universal Time (UTC) can be determined based on GPS code transfer and clock comparison [121]. Not only the GPS but also BeiDou can be synchronized with the clock of pseudolites via a similar method [122]. However, at present, the synchronization technology of BeiDou and groundbased pseudolites is still less researched. It is necessary to propose new synchronization schemes for BeiDou and pseudolites from a software perspective for common users who cannot deploy fiber optics in a large project.

3.3. Orbital and Ranging Problems

For pseudolites, the loss of positioning accuracy due to orbital errors is significant [123]. In open outdoor environments, users can receive the orbit information of satellites and even extrapolate the satellite orbits [124–126]. While indoors, the pseudolite orbits are fixed points. This orbit inconsistency problem between stationary pseudolites and dynamic satellites prevents indoor users from using the same method of positioning as outdoors. Unlike the near–far effect and clock synchronization problems that have been focused on in the past, there is little research related to the orbit problem of pseudolites in indoor positioning.

The solutions can be divided into two categories: one option is to disregard the outdoor satellite motion and simply treat pseudolites as technology similar to other indoor positioning technologies; although high-precision positioning can be achieved, this essentially eliminates the primary advantages of pseudolites, i.e., the seamless integration of indoor and outdoor location services and the consistency of GNSS and pseudolite spatial data; another option is to use pseudolites as both GNSS receivers and transmitters [127], replicating and forwarding ephemeris and range values; when GNSS ranging errors such as ionospheric delays, tropospheric delays, clock differences, orbital errors, antenna phase center corrections, differential code biases, tide corrections, solid tide corrections, relativistic effects, and multipath effects are adequately handled, pseudo-satellites can be used as receivers to achieve millimeter-level positioning accuracy. This approach implies that the user can receive the ephemeris and that the receiver's position can be unified under the GNSS framework. However, the range values received by the user are still provided by pseudolites and are not related to the real ephemeris. Therefore, the user's positioning is still based on the indoor datum and is not included in the GNSS framework, as shown in Figure 5.



Figure 5. The dynamic orbit of real satellites and the stationary orbit of pseudolites.

When completing the integrated spatial data construction of outdoor and indoor/ overground and underground activities with pseudolites, the orbit problem is unavoidable. A feasible method is to establish a two-way communication relationship between the user and the pseudolite transmitter based on the second option in the previous paragraph. The indoor positioning range values, altitude, and azimuth are returned to the pseudolite in real-time. Subsequently, the pseudolite calculates the range, altitude, and azimuth of the user and the real satellite based on its position and broadcasts them directionally.

3.4. Constellation Optimization

In addition to the method of receiver signal processing, the constellation optimization of the pseudolite is crucial to enabling sustainable signal reception and low dilution of precision (DOP). A study has identified the concept that the constellation optimization of pseudolites can influence the pseudolite positioning error [123]. The essence of constellation optimization is to make a larger number of satellites observable in more space to improve positioning precision [124,128]. The effectiveness of a multi-objective particle swarm-based approach for the deployment optimization of pseudolite systems is discussed [129]. Both theoretical and numerical analyses have shown [123] that the positioning error can be significantly reduced when the pseudolite is in an optimal position.

Moreover, the pseudolite constellation can be made more reasonable by testing the positioning accuracy of different locations of the receiver [130]. The geometric design of the pseudolite system can be determined via visual domain analysis and signal coverage time and base station accuracy factors [131].

Parameters related to the pseudolite constellation are the location of the pseudolites, the geometric space, and the obstruction condition of the signal path. An ideal constellation optimization method is to reconstruct the interior in three dimensions via visual or laser sensors beforehand. Then, the pseudolite layout that can obtain the maximum user-pseudolite viewing space is selected experimentally. Subsequently, the simulated configuration is corrected by evaluating the actual signal reception test.

4. Algorithmic and Technical Issues on the User Side

In addition to the system architecture and challenges, there are still problems to be solved via processing at signal receivers, such as poor geometry, large linearization errors, and signal reflection that cause multipath errors, leading to signal divergence. In addition, time delay fluctuations, phase center errors, and potential multipath errors disrupt the integer characteristics of the carrier phase [114]. To address these deficiencies, some algorithms can be used on the signal-receiving terminal.

4.1. Multipath

The multipath effect is the key to improving the accuracy of pseudolite positioning. The generation mechanism of the multipath effect in pseudo-satellite positioning is shown in Figure 6. In an experiment, 94% of the pseudolite positioning errors could be below 2 m when the errors introduced by multipath effects were controlled to be below 1 m [132]. In addition to a more sensible antenna design at the signal generation end, for example, two-array pseudolite antennas make the signal establish a long-distance link [133,134]. The ability to detect multiple paths at the receiver is another way to mitigate this error; for example, an area-based approach can detect more than 70% of the signal at low carrier-to-noise ratios [135].

From engineering experiments, a hardware–software complex is considered for studying the accuracy and noise immunity characteristics of pseudolite-based proximity navigation systems [136]. This complex is implemented based on the National Instruments hardware platform and the LabView coding environment. It provides a simulated navigation field and analyzes the received signals, such as signal transmission path loss and power angle distribution, which can be obtained by suppressing a ray-tracing method [137]. It also determines the error in the measurement of navigation parameters of pseudolite sig-



nals. Subsequently, the measured errors are compared with the characteristics of standard GNSS receivers.

Figure 6. Multipath effects in pseudo-satellite positioning: receivers receive both direct and reflected signals at the same time.

The reduction in available observations due to barriers in the signal path exacerbates the results of multipath errors. Non-visual range cases and under-observation can be handled using a machine-learning approach [138]. In solving the problem of line-of-sight between different transmitter-receiver pairs that differs greatly while performing antenna calibration, the multipath error is severe when there is no expensive anechoic chamber. Multiple calibrations at different locations can reduce antenna phase pattern accuracy to the millimeter level due to protection from multipath error [139]. An indoor fusion localization method using the double-difference pseudo-range measurements and the pseudolite signal carrier-to-noise ratio [140] demonstrates good suppression of receiver sampling time asynchrony and multipath errors. Challenges such as the poor geometric configuration of indoor pseudolites, the large linearization error, and the multipath error can be solved by the indoor PPP algorithm with optimal floating point ambiguity in the trust region [114].

The multipath effect depends on the location and constellation of the pseudolite. In addition to simply using signal processing methods to mitigate the multipath effect, the effect of the constellation should be considered when hardware conditions are limited. Furthermore, the multipath effect is related to indoor scenes. The user can assess the upcoming scene conditions by simultaneous localization and mapping (SLAM) or visual odometry (VO) to mitigate multipath errors [141] or to estimate the upcoming line-of-sight signal and multipath errors in advance to reduce the weight of pseudolite positioning in integrated navigation.

4.2. Resilient Processing Models

The resilient model contains the function model and the stochastic model [142], which are reviewed as follows.

4.2.1. The Function Model in Positioning

The function model expressed here is the process of modeling and calculating the user's location from the data received by the receiver side. The positioning of the pseudolite can be achieved via space resection, such as via the GNSS. In general, for low-cost single-frequency pseudolite systems, the static differential pseudolite system (DPL) method [143,144] is used to quickly acquire the low-precision positioning coordinates of the roaming station. A pseudolite positioning method using the carrier phase difference [145]

performs a single differential (SD) operation on the two sets of carrier phase measurements output from a dual-antenna receiver to eliminate the effect caused by the common error.

The ambiguity function method (AFM) was used to find coordinates in the corresponding ephemeris [144]. Confronting the non-linearity problem in the pseudolite system, the projection cancellation technique [146] is developed for linearizing the pseudo-range observations on the reference virtual station and user stations; the algorithm can improve the overall positioning accuracy by approximately 30% without increasing the time cost in the case of severe non-linearity. The optimal geometric factor method [147] based on the moving vector component and the multiarray pseudolite weighting factor for indoor positioning is proposed to solve problems such as single-array pseudolites and antenna aging or damage, and simulation results show that it effectively expands the high-precision positioning area. A method [144] is proposed to achieve high-precision indoor pseudolite positioning without using known point initialization (KPI), which can quickly obtain the initial coordinates and solve the ambiguity [148]. By classifying clock deviations as transmitter phase biases (TPBs) and estimating them, pseudolites can provide GNSS-like PPP-RTK services [149]. Above all, due to the similarity with the GNSS, the function model of pseudolite positioning can be borrowed from the GNSS models of several types and made adaptive modifications.

4.2.2. The Stochastic Model in Positioning

Kalman filtering and its adaptation for pseudolite positioning improvement have been studied. Although Kalman filtering can also be classified as a functional model [150], it is found that most of the pseudolite positioning methods related to error processing and gross error detection under the stochastic model are based on Kalman filtering, so it is discussed in this section. The extended Kalman filter (EKF) can be used for parameter estimation provided that the pseudolite operates properly and the number meets the positioning requirements [151]. Dynamic centimeter-level accuracy and static millimeter-level accuracy can be guaranteed through fault detection and troubleshooting. A non-linear parameter estimation method suitable for pseudolite localization [143], namely, the unscented Kalman filter (UKF), is introduced, which has high accuracy and low dependence on the initial coordinate values; the numerical results show that the computational efficiency of the EKF and UKF are basically the same, with the EKF performing slightly better. In response to the lack of resistance of the standard UKF to frequent anomalous observations, a robust UKF and partial ambiguity resolution (AR) algorithm [152] was proposed, which has the advantage of being more resistant to gross errors. A distributed algorithm based on the split covariance intersection filter (SCIF) is proposed [153] to process the newly acquired data to obtain consistently estimated states. In addition, different orbit and clock-difference correction prediction models are discussed. The performance of airbornepseudolite (A-PL) positioning based on GNSS PPP and inter-pseudolite ranges, as well as error prediction modeling, are evaluated. The results show that GNSS PPP combined with inter-pseudolites can improve convergence localization accuracy and shorten the convergence time of GNSS PPP.

4.2.3. Next Steps

Most existing experiments show that with an adequate number of pseudolites, accomplished clock synchronization, optimized constellation, and well-handled multipath errors, although the processing methods of signals or received data are different, the positioning accuracy of dynamic centimeter and static millimeter levels can be largely achieved. In the case of bad conditions, the positioning accuracy is at the meter level. In other words, the initial theory and methods of pseudolite positioning for indoor use have been developed. The current and future research trends can be summarized via three aspects. First, new ideas for pseudolite localization in the function model, for example, Doppler observation quantities, can be used for localization [154]. Additionally, by collecting the immediate and historical carrier phase data in the indoor environment and using a training model to predict the location of the positioning terminal, higher positioning accuracy can be achieved. An indoor localization system based on homologous array pseudolite database matching could be another method [155]. Second, in new application scenarios, new problems of pseudolite positioning would appear and be solved, for example, the detection and processing of coarse differences in a stochastic model in the case of signal jamming and the dynamic localization of indoor unmanned vehicles or drones using pseudolites. A third topic is the error model and processing method of deep integration of pseudolites and the GNSS, i.e., the integration of indoor and outdoor spatiotemporal data and seamless navigation tests. This would involve specific issues such as the switching of signal channels, unification of coordinate frames, etc.

4.3. Integrated Positioning with Other Methods

Multiple sources of sensors can be used in conjunction to obtain high-accuracy positioning with advantage sharing. The positioning accuracy of the GNSS integrated with pseudolites in an obscured environment was evaluated; integrated with PPP and pseudolites, submeter-level positioning is achievable in urban canyons [156]. As one of the most common indoor positioning techniques, integrated navigation with inertial guidance is mostly discussed. The principle simulation and testing of the combination of pseudolite, GPS, and inertial navigation system (INS) platforms have been analyzed [157]. The introduction of pseudolites into existing integrated GPS/INS systems to provide higher availability, integrity, and accuracy in localized areas has been discussed [158]. Preliminary experimental results on the dynamics of a combination of pseudolite, GPS, and INS platforms have been explored from an algorithmic perspective, and the results suggest that the problem of high noise of pseudolite signals and data utilization are of concern [159]. Combining the inertial measurement unit (IMU) in a test environment can improve the pseudolite positioning accuracy from 25 cm to 18 cm [160]. By combining GPS, IMU, pseudolite, and frontal laser scanner technologies, centimeter-level positioning accuracy can be obtained in an obscured environment [161].

Furthermore, studies show that integration with an INS can assist pseudolites in measuring ambiguity [162]. A suitable stochastic model allows pseudolites to help GPS ambiguity resolution, especially when the line-of-sight vector is changing rapidly [163]. The cycle slip detection of pseudolites can be assisted by using a low-cost IMU [164]. An ultra-tight integrated model incorporating GPS, INS, and pseudolite was explored [165]. The application scope of an integrated navigation system incorporating pseudolite, GPS, and INS technologies could include civil aviation [29].

The combination of pseudolite and pedestrian dead recycling (PDR) is another approach that has been studied. The average positioning error of PDR is 5.5 m and 10 m, while the average positioning error of the integrated positioning is 2–3 m [166]. The results above show that the integrated technique of pseudolite and PDR can achieve high-accuracy positioning over a long time and long distance. In addition to PDR, synthetic marks can be taken into account [85]. Alternatively, other indoor positioning techniques can be integrated with pseudolites; for example, the performance of pseudolites and UWB transmission was evaluated [167]. Magnetometers [168] can also be included in the category of techniques for assisted positioning.

Although they are different systems and positioning principles, for the GNSS or pseudolites, inertial devices, PDR, and other methods can provide technical support for outdoor indoor and aboveground underground continuous navigation.

In an enlightening manner, it was demonstrated that SLAM can be used to assist in the construction of a dedicated pseudolite system to explore transmitter asynchrony [169]. A pseudolite multi-source fusion navigation and positioning algorithm with map information constraints was proposed to solve the localization problem in narrow indoor areas [170]; that is, the direct arrival signal of a single pseudolite can be used to correct the mobile phone position information under map constraints in order to improve the navigation and positioning performance in indoor access and corridor areas. In addition, an integrated

visual–inertial odometry (VIO) and pseudolite navigation scheme was proposed [141]. However, few studies have been conducted to explore the integrated navigation of pseudolites and SLAM. SLAM allows for fast indoor modeling to obtain the locations of pseudolites. This is important for users who do not know the constellation and proximity of pseudolites to deal with the multipath and near–far effect. The fusion and application of the GNSS, pseudolites, and SLAM-containing semantics can be an important support in the framework of a smart city or meta-universe.

GNSS or pseudo-satellites can provide positions in the ECEF framework, but with low or failed positioning accuracy when the received signal quality is poor. Indoor positioning technologies such as UWB, while providing highly accurate positions, have a self-defined coordinate system that is not geo-referenced; therefore, a possible option is for GNSS to provide a rough absolute geographic position via pseudo-satellites, and later combine it with a variety of other techniques to achieve more accurate positioning, as shown in Figure 7.



Figure 7. One possible solution: GNSS provides a rough absolute location via pseudolites, which is later integrated with a variety of other technologies to achieve more accurate positioning.

5. Pseudolites as a Spatial Reference: Discussion and Prospects

A unified, stable spatial foundation is a prerequisite for smart city construction. Nowadays, GNSS-based spatial data are available on the ground, in the case of China, for example, the "China Geodetic Coordinate System 2000" (CGCS2000) was built [171] and maintained [172,173], and the "Seafloor Geodetic Network" [174] has been gradually carried out. However, at present, industrial and subsistence activities are gradually entering the underground space, such as the mega underground construction in the Xiong'an New Area [175] and the underground high-speed railway station of Daxing Airport [176] near Beijing. The indoor/underground spatial datum and its location-based services are becoming a rigid demand. As the technical problems in pseudolites are gradually solved, it is expected to become part of the space infrastructure, as shown in Figure 8. In particular, if the orbits and clocks of the pseudo-satellites are harmonized with the GNSS, it will provide users with GNSS-like spatiotemporal information, and then location-based IoT services such as underground or indoor navigation, autonomous driving, and so on, will greatly benefit.

Regarding strict national space reference requirements and the daily needs of mass users, new problems have arisen when using pseudolites as a space reference and seamless navigation method as well as an indoor positioning method. Among them, the most important is the precise correlation of references, e.g., indoor and outdoor references. This requires that time synchronization and orbit issues be addressed in the framework of GNSS positioning and timing. For specific industries, such as the IoT and autonomous driving, pseudolite positioning solutions need to interface with pre-existing industry standards,



side and the user side sorted out above, some potentially useful suggestions or discussions

have been organized as follows.

Figure 8. Key technologies and applications for pseudolites: datum unification is the intermediate link for selected applications.

From the hardware perspective, basal signal emission and capture are largely solved problems, from which the most serious and most explored is the problem of near–far effects. The near–far effect that occurs when multiple signals are received simultaneously is presented as a fundamental problem to be addressed in various ways [83,95,177–181], with most of them needing to improve or increase the signal reception equipment and mode. However, with the calling of seamless navigation, the near–far effect problem needs to be mitigated while using the most common receivers. It is worth noting that although the problems in pseudolites are divided in the above subsection, many of them need to be addressed comprehensively. The near–far effect is closely related to the pseudolite constellation. In constellation design, the PVT (position, velocity, and time) [180] as a function of the near–far effect could be evaluated. Afterward, from a geometric viewpoint, an optimal constellation could be designed [22] to maximize the number of pseudolite observations at more locations and minimize the impact of the near–far effect and multipath effect. In this process, signal transponders can be used to improve signal strength in some of the priority areas.

New test methods can be used in the process of constellation design. For example, visual sensors [182] can quantitatively sense the proximity of a pseudolite. As the sensor moves, the distance and proximity effects of the signal change. The receiver–pseudolite distance in a three-dimensional space thus constructed is related as a function of the near-far effect and the geometric accuracy factor [22,130,183], which helps in the optimization of the constellation.

The visual 3D space construction is meaningful for indoor integrated navigation [42]; it is more important to allow the user to evaluate, or even perceive, the multipath in advance. The relationship between distance and near–far effects as a function of distance can be incorporated into the algorithm of integrated navigation [184], giving more weight to the

navigation of other methods when encountering near–far effects and multipath. Multiple optimization methods of intelligent PNT [185–187] can be incorporated into the integrated navigation of pseudolites.

The ranging function of the pseudolite can be used to synchronize time [23]. Each pseudolite can act as both a transmitter and a receiver. When the constellation is determined and the distance between every two pseudolites is known, the clock deviation between the pseudolites can be determined by dividing their distance measurements from each other by the speed of light.

In addition to being a positioning technique, the transfer of the GNSS datum is an important task that should be undertaken by pseudolites in seamless navigation [188]. For space–time uniform PNT or seamless navigation, there are two suggested options. One is to first set an indoor local datum, such as the north–east–up local coordinate system in Figure 9. The pseudolite acts as a GNSS positioning terminal and also as a transmitter; thus, it has different coordinates in both the WGS84 coordinate system and the local coordinate system. After that, multiple such pseudolites are used as intermediaries to complete the 3D coordinate conversion and thus transfer the user coordinates to the WGS84 frame [189]. The other option is to broadcast the satellite and user range values directly via pseudolites. In Figure 9, since the two range values, p_1 and p_2 , are known, the angle between p_1 and p_2 can be calculated when the altitude angle of the outdoor pseudolite and the altitude angle of the indoor user are known. It is worth noting that this calculation needs to be performed in the plane determined by 1 and 2. The GNSS satellite-to-user range value p_3 can be calculated according to the law of cosines.



Figure 9. A schematic diagram of pseudolite-mediated frame unification.

From specific experimental scenarios, concise solutions for the practical needs of users for integrated indoor and outdoor navigation should be highlighted more often. The solution to these issues will help pseudolites move toward engineering applications to become the underlying technology of the IoT [190–192], as well as to serve the construction of new national data. Even further, as shown in Figure 8, in the future of unmanned driving, pseudolites could provide urban canyon navigation and underground garage location services and could also offer human–vehicle interconnection with the GNSS as the datum [193].

6. Pseudolite-Based Location Service: Conceived Application in Smart Cities

In the above text, some traditional application scenarios such as augmented navigation of airplanes, trains, ships, and especially indoor navigation were presented. In today's smart city context, these traditional applications can also be included in the smart city scope. In addition, there are the following predefined scenarios for pseudo-satellite-based localization to play a role.

6.1. Intelligent Transportation

Traffic congestion is one of the important problems being addressed by smart cities. Pseudolite-based location services can provide precise location information for all vehicles including those in GNSS-denied environments. In turn, it can be used by managers to obtain real-time traffic congestion conditions. In this way, scheduling is conducted to maximize the efficiency of the traffic.

Intelligent driving is developing rapidly. In smart driving, vision-based, laser-based, etc., positioning methods only allow a vehicle to obtain a position in a localized coordinate system, whereas pseudo-satellites can provide the vehicle with an absolute position. This can match vehicles with electronic maps and improve the accuracy, integrity, continuity, and availability of location services for autonomous driving.

6.2. Smart Health and Lifestyle

Pseudolite-based positioning technology can bring positive results for smart health. In outdoor exercise, people wear bracelets, watches, etc., that can count steps in various ways such as through the GNSS. However, indoors, inaccurate positioning often leads to inaccurate recording results. Pseudolite-based localization would help to improve the accuracy of exercise recordings while assisting in location-based monitoring for medical treatment.

People often become lost in large buildings such as shopping malls and airports. Traditional indoor localization only tells the user "where you are" but not "where you are going". Pseudo-satellites will help interconnect the location between people and people, between things and things, and between people and things, updating people's knowledge of space and time and their way of life.

6.3. Public Safety and Emergency Response

Public safety is an important aspect of location-based services. Law enforcement agencies can use pseudo-satellite-based location data to monitor possible criminal activities and query historical location trajectories to obtain evidence.

In addition, pseudo-satellite-based location services can assist in providing early warning of earthquakes. Although some cell phones currently have the capability to report earthquakes, their false alarm rate is high. Pseudo-satellite bodies and users are sensitive to location, and pseudo-satellites have the ability to naturally monitor building vibration. Therefore, based on the location information of the pseudo-satellite, seismic reporting capabilities will be enhanced.

7. Conclusions

This paper introduces pseudolite as a useful tool from the perspective of integrated location services in smart cities. First, the concept and development history of pseudolites were introduced; second, the possible application scenarios of pseudolites in smart cities were presented; after that, the problems on the system side of pseudolites and the methods of handling them on the user side were reviewed and analyzed; then, with the aim of applying pseudolites to smart cities, we discuss some solutions and application ideas for tackling the problems of pseudolites.

With the development of pseudolite technology over the years, indoor positioning based on pseudolites is already possible from the perspective of scientific research, especially as the problems of time synchronization, the near–far effects of signals, and multipath errors have been largely solved or mitigated. Pseudolites have already shown their application prospects with respect to smart cities. With pseudolite devices having been developed, innovation in the process of practical test applications is the next point of focus. In this process, specific issues such as constellation optimization need to be considered. Some of our ideas are summarized as follows:

On the system side: (1) facing the problem of near–far effects, the mode and intensity of transmitting signals and the hardware design of receiving devices and their anti-jamming capability should continue to be investigated; (2) for consumer-level indoor positioning,

fiber optic synchronization can be adopted to replace expensive atomic clocks; (3) for the unification of indoor and outdoor coordinate frames, algorithms or methods to accurately correlate the orbit and ranging data of pseudolites with GNSS satellites should be explored; (4) for the problem of user positioning accuracy, the placement of indoor pseudo-satellites should be made more reasonable so as to allow users to obtain redundant observations to participate in the adjustment.

On the user side: (1) in indoor or underground scenarios, multipath effects seem to be unavoidable, but it is possible to study how to distinguish between direct and reflected signals, or even use reflected signals for positioning; (2) involving the processing of userside data and fusion with other sensors in addition to positioning accuracy, the resilience, availability, and integrity of a single system or multiple systems should also be explored in the future.

We believe that as a provider of location references, indoor or underground positioning services based on pseudolites may contribute to smart cities as much as GNSS-based services in open areas.

Author Contributions: Conceptualization, G.X.; methodology, G.X.; investigation, T.L., J.L., B.Z., Y.M. and M.S.; resources, J.W. and H.Z.; writing, T.L.; visualization, T.L.; supervision, Z.L.; project administration, Z.L.; funding acquisition, G.X. All authors have read and agreed to the published version of the manuscript.

Funding: This study was supported by National Key R&D Program of China (2022YFC3003800). This study is from the project supported by the open fund of key laboratory of urban land resources monitoring and simulation, Ministry of Natural Resources (KF-2021-06-104).

Data Availability Statement: This review contains no data.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Minetto, A.; Dovis, F.; Vesco, A.; Garcia-Fernandez, M.; López-Cruces, A.; Trigo, J.L.; Molina, M.; Pérez-Conesa, A.; Gáñez-Fernández, J.; Seco-Granados, G. A testbed for GNSS-based positioning and navigation technologies in smart cities: The HANSEL project. *Smart Cities* 2020, 3, 1219–1241. [CrossRef]
- van Den Bossche, A.; Dalcé, R.; Val, T. LocURa4IoT—A Testbed Dedicated to Accurate Localization of Wireless Nodes in the IoT. IEEE Sens. J. 2021, 22, 5437–5446. [CrossRef]
- Yang, B.; Guo, L.; Guo, R.; Zhao, M.; Zhao, T. A novel trilateration algorithm for RSSI-based indoor localization. *IEEE Sens. J.* 2020, 20, 8164–8172. [CrossRef]
- Wang, W.; Zhu, Q.; Wang, Z.; Zhao, X.; Yang, Y. Research on Indoor Positioning Algorithm Based on SAGA-BP Neural Network. IEEE Sens. J. 2021, 22, 3736–3744. [CrossRef]
- Blewitt, G.; Boucher, C.; Davies, P.; Heflin, M.; Herring, T.; Kouba, J. ITRF Densification and Continuous Realization by the IGS. In *Advances in Positioning and Reference Frames*; Springer: Berlin/Heidelberg, Germany, 1998; pp. 8–17.
- 6. El-Sheimy, N.; Li, Y. Indoor navigation: State of the art and future trends. Satell. Navig. 2021, 2, 7. [CrossRef]
- Sonnessa, A.; Saponaro, M.; Alfio, V.S.; Capolupo, A.; Turso, A.; Tarantino, E. Indoor Positioning Methods—A Short Review and First Tests Using a Robotic Platform for Tunnel Monitoring. In Proceedings of the International Conference on Computational Science and Its Applications, Cagliari, Italy, 1–4 July 2020; pp. 664–679.
- 8. Yang, Y.; Mao, Y.; Sun, B. Basic performance and future developments of BeiDou global navigation satellite system. *Satellite Navigation* **2020**, *1*, 1. [CrossRef]
- Zuo, Z.-Y.; Qiao, X.; Wu, Y.-B. Concepts of comprehensive PNT and related key technologies. In Proceedings of the International Conference on Modeling, Analysis, Simulation Technologies and Applications, Hangzhou, China, 26–27 May 2019.
- Kim, C.; So, H.; Lee, T.; Kee, C. A pseudolite-based positioning system for legacy GNSS receivers. Sensors 2014, 14, 6104–6123. [CrossRef]
- 11. Wang, J. Pseudolite applications in positioning and navigation: Pro-gress and problems. J. Glob. Position. Syst. 2002, 1, 8–56. [CrossRef]
- 12. Harrington, R.; Dolloff, J. The inverted range: GPS user test facility. In Proceedings of the IEEE PLANS'76, San Diego, CA, USA, 1–2 November 1976; pp. 204–211.
- Cobb, H.S.; Cohen, C.E.; Parkinson, B.W. Theory and design of pseudolites. In Proceedings of the 1994 National Technical Meeting of The Institute of Navigation, San Diego, CA, USA, 24–26 January 1994; pp. 69–75.
- 14. Cobb, H.S. GPS Pseudolites: Theory, Design, and Applications; Stanford University: Stanford, CA, USA, 1997.

- 15. Beser, J.; Parkinson, B.W. The application of NAVSTAR differential GPS in the civilian community. *Navig. J. Inst. Navig.* **1982**, 29, 107–136. [CrossRef]
- Kalafus, R.M.; Van Dierendonck, A.; Pealer, N.A. Special Committee 104 recommendations for differential GPS service. In Proceedings of the 42nd Annual Meeting of the Institute of Navigation, Seattle, DC, USA, 24–26 June 1986; pp. 45–54.
- Wolf, R.; Thalhammer, M.; Hein, G.W. GATE-The german galileo test environment. In Proceedings of the 16th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS/GNSS 2003), Portland, OR, USA, 9–12 September 2003; pp. 1009–1015.
- Kuusniemi, H.; Bhuiyan, M.Z.H.; Ström, M.; Söderholm, S.; Jokitalo, T.; Chen, L.; Chen, R. Utilizing pulsed pseudolites and high-sensitivity GNSS for ubiquitous outdoor/indoor satellite navigation. In Proceedings of the 2012 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Sydney, NSW, Australia, 13–15 November 2012; pp. 1–7.
- 19. Chen, R.; Hyttinen, A.; Chen, Y.; Strom, M.; Laitinen, H.; Tossaint, M.; Martin, S. Development of the EGNOS pseudolite system. *J. Glob. Position. Syst.* **2007**, *6*, 119–125. [CrossRef]
- Montefusco, C.; Traveset, J.; Rodriguez, R.; Toran, F.; Jofre, M. Enhancing SBAS Performance: The EGNOS pseudolite concept. In Proceedings of the European Navigation Conference, Munich, Germany, 19–22 July 2005.
- Kang, G.; Tan, L.; Hua, B.; Zheng, F. Study on p seudolite system for BeiDou based on dynamic and independent aircrafts configuration. In Proceedings of the China Satellite Navigation Conference (CSNC) 2013 Proceedings, Wuhan, China, 15–17 May 2013; pp. 159–172.
- Chao, M.; Jinling, W.; Jianyun, C. Beidou compatible indoor positioning system architecture design and research on geometry of pseudolite. In Proceedings of the 2016 Fourth International Conference on Ubiquitous Positioning, Indoor Navigation and Location Based Services (UPINLBS), Shanghai, China, 2–4 November 2016; pp. 176–181.
- Gan, X.; Yu, B.; Chao, L.; Liu, S. The development, test and application of new technology on Beidou/GPS dual-mode pseudolites. In Proceedings of the China Satellite Navigation Conference (CSNC) 2015 Proceedings, Xi'an, China, 13–15 May 2015; Volume I, pp. 353–364.
- Hein, G.W.; Eissfeller, B.; Werner, W.; Ott, B.; Elrod, B.D.; Barltrop, K.; Stafford, J. Practical investigations on DGPS for aircraft precision approaches augmented by pseudolite carrier phase tracking. In Proceedings of the 10th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS 1997), Kansas City, MO, USA, 16–19 September 1997; pp. 1851–1860.
- Cohen, C.E. Real-time cycle ambiguity resolution using a pseudolite for precision landing of aircraft with GPS. In Proceedings of the 2nd DSNS, Amsterdam, The Netherlands, 30 March–2 April 1991.
- Elrod, B.; Barltrop, K.; Van Dierendonck, A. Testing of GPS augmented with pseudolites for precision approach applications. In Proceedings of the 7th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS 1994), Salt Lake City, UT, USA, 20–23 September 1994; pp. 1269–1278.
- Van Dierendonck, A.; Fenton, P.; Hegarty, C. Proposed airport pseudolite signal specification for GPS precision approach local area augmentation systems. In Proceedings of the 10th International Technical Meeting of the Satellite Division of the Institute of Navigation (ION GPS 1997), Kansas City, MO, USA, 16–19 September 1997; pp. 1603–1612.
- 28. Ko, P.-Y. GPS-Based Precision Approach and Landing Navigation: Emphasis on Inertial and Pseudolite Augmentation and Differential Ionosphere Effect. Master's Thesis, Stanford University, Stanford, CA, USA, 2000.
- 29. Lee, H.-K.; Soon, B.; Barnes, J.; Wang, J.; Rizos, C. Experimental analysis of GPS/Pseudolite/INS integration for aircraft precision approach and landing. *J. Navig.* 2008, *61*, 257–270. [CrossRef]
- 30. Pervan, B.S.; Cohen, C.E.; Parkinson, B.W. Integrity monitoring for precision approach using kinematic GPS and a ground-based pseudolite. *J. Inst. Navig.* **1994**, *41*, 159–174. [CrossRef]
- Galijan, R.C.; Lucha, G.V. A suggested approach for augmenting GNSS category III approaches and landings: The GPS/Glonass and Glonass pseudolite system. In Proceedings of the 6th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS 1993), Salt Lake City, UT, USA, 22–24 September 1993; pp. 157–160.
- 32. Yun, H.; Han, D.; Kim, C.; Kim, O.-J.; Song, J.; No, H.; Kim, J.; Kim, D.; Kee, C. Development of Pseudolite Based Augmentation System (PBAS). In Proceedings of the ION 2015 Pacific PNT Meeting, Honolulu, HI, USA, 20–23 April 2015; pp. 1055–1058.
- Yun, H.; Han, D.; Kee, C. Preliminary Test Results of Pseudolite-Based Augmentation System (PBAS). In Proceedings of the ION 2013 Pacific PNT Meeting, Honolulu, HI, USA, 23–25 April 2013; pp. 631–635.
- Amt, J.H.; Raquet, J.F. Flight testing of a pseudolite navigation system on a UAV. In Proceedings of the 2007 National Technical Meeting of the Institute of Navigation, San Diego, CA, USA, 22–24 January 2007; pp. 1147–1154.
- Lee, H.K.; Wang, J.; Rizos, C.; Barnes, J. Analysis of pseudolite augmentation for GPS airborne applications. In Proceedings of the 15th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS 2002), Portland, OR, USA, 24–27 September 2002; pp. 2610–2618.
- Wang, J.J. Integration of GPS, INS and Pseudolite to Geo-Reference Surveying and Mapping Systems. Ph.D. Thesis, UNSW Sydney, Sydney, NSW, Australia, 2007.
- 37. Skrypnik, O.; Kargapol'cev, S.; Sizykh, V.; Daneev, A.; Aref'ev, R. Characteristics of the integrated GLONASS accuracy field in the optimal placement of pseudo satellites in the aerodrome zone. *Appl. Anal. Discret. Math.* **2018**, *19*, 83–91. [CrossRef]
- Lo, S.; Chen, Y.; Enge, P.; Pelgrum, W.; Soelter, A. Flight test of a pseudo-ranging signal compatible with existing distance measuring equipment (DME) ground stations. *Navigation* 2020, 67, 567–582. [CrossRef]

- 39. Ma, C.; Yang, J.; Chen, J.; Tang, Y. Time synchronization requirement of global navigation satellite system augmentation system based on pseudolite. *Meas. Control* 2019, *52*, 303–313. [CrossRef]
- Ma, W.; Yuan, J.; Luo, J. Airborne pseudolite aiding BeiDou system to improve positioning precision in low latitude areas. In Proceedings of the International Conference on Space Information Technology, Wuhan, China, 19–20 November 2005. [CrossRef]
- Zhao, X.; Liu, J.; Cai, B.; Lu, D.; Wang, J. Research on Optimized Pseudolite Constellation Design under Constrained GNSS Environment in Railway Stations. In Proceedings of the 2019 IEEE Intelligent Transportation Systems Conference—ITSC, Auckland, New Zealand, 27–30 October 2019.
- 42. Liu, J.; Zhao, X.-L.; Cai, B.-G.; Wang, J. Pseudolite Constellation Optimization for Seamless Train Positioning in GNSS-Challenged Railway Stations. *IEEE Trans. Intell. Transp. Syst.* **2021**, 23, 13636–13654. [CrossRef]
- 43. Progri, I.F.; Michalson, W.R. An underground system of stationary pseudolites. J. Geolocation Geo-Inf. Geo-Intell. 2020, 2020, 38–47. [CrossRef]
- 44. Youngberg, J.W. A novel method for extending GPS to underwater applications. Navigation 1991, 38, 263–271. [CrossRef]
- 45. Tiwary, K.; Sharada, G.; Singh, A. Underwater Navigation using Pseudolite. Def. Sci. J. 2011, 61, 331–336. [CrossRef]
- Yang, Y.; Xu, T.; Xue, S. Progresses and prospects of marine geodetic datum and marine navigation in China. J. Geod. Geoinf. Sci. 2019, 1, 16–24.
- 47. Yang, Y.; Qin, X. Resilient observation models for seafloor geodetic positioning. J. Geod. 2021, 95, 79. [CrossRef]
- Grosch, A.; Enneking, C.; Greda, L.A.; Tanajewski, D.; Grunwald, G.; Ciećko, A. Theoretical Concept for a Mobile Underwater Radio-Navigation System Using Pseudolite Buoys. *Remote Sens.* 2020, *12*, 3636. [CrossRef]
- Qi, K.; Qu, G.; Xue, S.; Xu, T.; Su, X.; Liu, Y.; Wan, J. Analytical optimization on GNSS buoy array for underwater positioning. *Acta Oceanol. Sin.* 2019, 38, 137–143. [CrossRef]
- Dai, K.; Li, Z.; Xu, Q.; Bürgmann, R.; Milledge, D.G.; Tomas, R.; Fan, X.; Zhao, C.; Liu, X.; Peng, J. Entering the era of earth observation-based landslide warning systems: A novel and exciting framework. *IEEE Geosci. Remote Sens. Mag.* 2020, *8*, 136–153. [CrossRef]
- 51. Benoit, L.; Briole, P.; Martin, O.; Thom, C.; Malet, J.-P.; Ulrich, P. Monitoring landslide displacements with the Geocube wireless network of low-cost GPS. *Eng. Geol.* **2015**, *195*, 111–121. [CrossRef]
- 52. Shen, N.; Chen, L.; Wang, L.; Hu, H.; Lu, X.; Qian, C.; Liu, J.; Jin, S.; Chen, R. Short-term landslide displacement detection based on GNSS real-time kinematic positioning. *IEEE Trans. Instrum. Meas.* **2021**, *70*, 1004714. [CrossRef]
- 53. He, Y.; Xu, G.; Kaufmann, H.; Wang, J.; Ma, H.; Liu, T. Integration of InSAR and LiDAR Technologies for a Detailed Urban Subsidence and Hazard Assessment in Shenzhen, China. *Remote Sens.* **2021**, *13*, 2366. [CrossRef]
- 54. Hu, B.; Chen, J.; Zhang, X. Monitoring the land subsidence area in a coastal urban area with InSAR and GNSS. *Sensors* **2019**, *19*, 3181. [CrossRef]
- 55. Liu, C.; Gao, J.; Zhao, X.; Zhang, A.; Yu, X. Simulation and experiment analysis of dynamic deformation monitoring with the integrated GPS/pseudolite system. *J. Geophys. Eng.* **2014**, *12*, 45–56. [CrossRef]
- 56. Meng, X.; Roberts, G.; Dodson, A.; Cosser, E.; Barnes, J.; Rizos, C. Impact of GPS satellite and pseudolite geometry on structural deformation monitoring: Analytical and empirical studies. *J. Geod.* **2004**, *77*, 809–822. [CrossRef]
- 57. Scaioni, M.; Marsella, M.; Crosetto, M.; Tornatore, V.; Wang, J. Geodetic and remote-sensing sensors for dam deformation monitoring. *Sensors* 2018, 18, 3682. [CrossRef] [PubMed]
- Guoliang, S.; Lifang, W.; Rui, F.; Weihua, C. GNSS for Lunar Surface Positioning Based on Pseudo-satellites. In Proceedings of the 2019 IEEE 8th Joint International Information Technology and Artificial Intelligence Conference (ITAIC), Chongqing, China, 24–26 May 2019; pp. 1348–1353.
- LeMaster, E.A.; Rock, S.M. A local-area GPS pseudolite-based navigation system for mars rovers. *Auton. Robot.* 2003, 14, 209–224. [CrossRef]
- Schmitz, N.; Gao, Y.; Oberst, J.; Wild, A. A Moon-based Sensors Network supporting Science and Navigation. In Proceedings of the Geophysical Research Abstracts, Vienna, Austria, 13–18 April 2008.
- 61. Anzalone, E.; Iyer, A.; Statham, T. Use of Navigation Beacons to Support Lunar Vehicle Operations. In Proceedings of the 2020 IEEE Aerospace Conference, Big Sky, MT, USA, 7–14 March 2020; pp. 1–13.
- 62. Bonenberg, L.K.; Hancock, C.M.; Roberts, G.W. Indoor multipath effect study on the Locata system. J. Appl. Geod. 2010, 4, 137–143. [CrossRef]
- 63. Barnes, J.; Rizos, C.; Wang, J.; Small, D.; Voigt, G.; Gambale, N. Locata: The positioning technology of the future. In Proceedings of the 6th International Symposium on Satellite Navigation Technology Including Mobile Positioning & Location Services, Melbourne, VIC, Australia, 22–25 July 2003.
- 64. Bonenberg, L.; Roberts, G.; Hancock, C. Using Locata to augment GNSS in a kinematic urban environment. *Arch. Fotogram. Kartogr. I Teledetekcji* **2011**, 22, 63–74.
- 65. Roberts, G.W.; Bonenberg, L.K.; Hancock, C.M. Integrating Locatalites and GNSS for engineering works. In Proceedings of the 7th FIG Regional Conference, Hanoi, Vietnam, 19–22 October 2009.
- 66. Montillet, J.-P.; Roberts, G.; Hancock, C.; Meng, X.; Ogundipe, O.; Barnes, J. Deploying a Locata network to enable precise positioning in urban canyons. *J. Geod.* **2009**, *83*, 91–103. [CrossRef]
- 67. Bonenberg, L.K.; Hancock, C.; Roberts, G.W. Locata performance in a long term monitoring. J. Appl. Geod. 2013, 7, 271–280. [CrossRef]

- Montillet, J.-P.; Bonenberg, L.K.; Hancock, C.M.; Roberts, G.W. On the improvements of the single point positioning accuracy with Locata technology. GPS Solut. 2014, 18, 273–282. [CrossRef]
- 69. Jiang, W.; Li, Y.; Rizos, C. On-the-fly Locata/inertial navigation system integration for precise maritime application. *Meas. Sci. Technol.* **2013**, 24, 105104. [CrossRef]
- Bertsch, J.; Choudhury, M.; Rizos, C.; Kahle, H.-G. On-the-fly ambiguity resolution for Locata. In Proceedings of the International Symposium on GPS/GNSS, Gold Coast, QLD, Australia, 1–3 December 2009; pp. 1–3.
- Choudhury, M. Analysing Locata for Slow Structural Displacement Monitoring Application. Ph.D. Thesis, The University of New South Wales Sydney, Sydney, NSW, Australia, 2011.
- Rizos, C.; Yang, L. Background and recent advances in the locata terrestrial positioning and timing technology. *Sensors* 2019, 19, 1821. [CrossRef]
- Shockley, J.; Raquet, J. Estimation and mitigation of unmodeled errors for a pseudolite based reference system. In Proceedings of the 19th International Technical Meeting of the Satellite Division of the Institute of Navigation (ION GNSS 2006), Fort Worth, TX, USA, 26–29 September 2006; pp. 853–862.
- Barnes, J.; Rizos, C.; Kanli, M.; Pahwa, A. A solution to tough GNSS land applications using terrestrial-based transceivers (LocataLites). In Proceedings of the 19th International Technical Meeting of the Satellite Division of the Institute of Navigation (ION GNSS 2006), Fort Worth, TX, USA, 26–29 September 2006; pp. 1487–1493.
- Barnes, J.; Lamance, J.; Lilly, B.; Rogers, I.; Nix, M.; DeBeers, A. An integrated Locata & Leica Geosystems positioning system for open-cut mining applications. In Proceedings of the 20th International Technical Meeting of the Satellite Division of the Institute of Navigation (ION GNSS 2007), Fort Worth, TX, USA, 25–28 September 2007; pp. 1661–1668.
- 76. Barnes, J.; Rizos, C.; Pahwa, A.; Politi, N.; van Cranenbroeck, J. The potential of locata technology for structural monitoring applications. *Positioning* **2007**, *6*, 166–172. [CrossRef]
- Trunzo, A.; Benshoof, P.; Amt, J. The UHARS non-GPS based positioning system. In Proceedings of the 24th International Technical Meeting of The Satellite Division of the Institute of Navigation (ION GNSS 2011), Portland, OR, USA, 20–23 September 2011; pp. 3582–3586.
- Jiang, W.; Li, Y.; Rizos, C.; Barnes, J. Flight evaluation of a locata-augmented multisensor navigation system. J. Appl. Geod. 2013, 7, 281–290. [CrossRef]
- 79. Jiang, W.; Li, Y.; Rizos, C. Improved decentralized multi-sensor navigation system for airborne applications. *GPS Solut.* **2018**, 22, 78. [CrossRef]
- Yang, L. Performance of fault detection and exclusion for GNSS/Locata integrated navigation. In Proceedings of the 26th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS+ 2013), Nashville, TN, USA, 16–20 September 2013; pp. 1242–1251.
- Jiang, W.; Li, Y.; Rizos, C. A multisensor navigation system based on an adaptive fault-tolerant GOF algorithm. *IEEE Trans. Intell. Transp. Syst.* 2016, 18, 103–113. [CrossRef]
- He, C.; Yu, B. Signal Design of High Accuracy Terrestrial Pseudolites System in BeiDou RDSS Frequency Band. In Proceedings of the China Satellite Navigation Conference (CSNC) 2016 Proceedings, Changsha, China, 18–20 May 2016; Volume II, pp. 291–301.
 Hong-Jun, Y.E. Analysis and Research on the Near-far Effect of Pseudolites. *Radio Eng.* 2010, 40, 31–33.
- Fan, G.; Sheng, C.; Yu, B.; Huang, L.; Rong, Q. An Indoor and Outdoor Multi-Source Elastic Fusion Navigation and Positioning Algorithm Based on Particle Filters. *Future Internet* 2022, 14, 169. [CrossRef]
- 85. Zhu, R.; Wang, Y.; Cao, H.; Yu, B.; Gan, X.; Huang, L.; Zhang, H.; Li, S.; Jia, H.; Chen, J. RTK/pseudolite/LAHDE/IMU-PDR integrated pedestrian navigation system for urban and indoor environments. *Sensors* **2020**, *20*, 1791. [CrossRef]
- Li, B.; Zheng, J.; Fang, Y.; Yang, M.; Yan, Z. IoT as a Service. In Proceedings of the 5th EAI International Conference, IoTaaS 2019, Xi'an, China, 16–17 November 2019; Springer Nature: Berlin/Heidelberg, Germany, 2020; Volume 316.
- Li, Y.; Zhuang, Y.; Hu, X.; Gao, Z.; Hu, J.; Chen, L.; He, Z.; Pei, L.; Chen, K.; Wang, M. Toward location-enabled IoT (LE-IoT): IoT positioning techniques, error sources, and error mitigation. *IEEE Internet Things J.* 2020, *8*, 4035–4062. [CrossRef]
- 88. Nagarajan, S.G.; Zhang, P.; Nevat, I. Geo-spatial location estimation for Internet of Things (IoT) networks with one-way time-of-arrival via stochastic censoring. *IEEE Internet Things J.* **2016**, *4*, 205–214. [CrossRef]
- 89. Cao, Z.; Chen, R.; Guo, G.; Pan, Y. iBaby: A low cost BLE pseudolite based indoor baby care system. In Proceedings of the 2018 Ubiquitous Positioning, Indoor Navigation and Location-Based Services (UPINLBS), Wuhan, China, 22–23 March 2018; pp. 1–6.
- 90. Del Peral-Rosado, J.; Granados, G.; Raulefs, R.; Leitinger, E.; Grebien, S.; Wilding, T.; Dardari, D.; Lohan, E.; Wymeersch, H.; Floch, J. Whitepaper on New Localization Methods for 5G Wireless Systems and the Internet-of-Things. 2018. Available online: https://re.public.polimi.it/bitstream/11311/1069386/2/2018_white_paper_IRACON-WP2.pdf (accessed on 17 July 2023).
- 91. Tobie, A.-M.; Garcia-Pena, A.; Thevenon, P.; Vezinet, J.; Aubault, M. Hybrid navigation filters performances between GPS, Galileo and 5G TOA measurements in multipath environment. In Proceedings of the 33rd International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS+ 2020), Online, 21–25 September 2020; pp. 2107–2140.
- 92. Liu, J.; Gao, K.; Guo, W.; Cui, J.; Guo, C. Role, path, and vision of "5G+ BDS/GNSS". Satell. Navig. 2020, 1, 23. [CrossRef]
- 93. Ai, G.-X.; Shi, H.-L.; Wu, H.-T.; Yan, Y.-H.; Bian, Y.-J.; Hu, Y.-H.; Li, Z.-G.; Guo, J.; Cai, X.-D. A positioning system based on communication satellites and the Chinese Area Positioning System (CAPS). *Chin. J. Astron. Astrophys.* **2008**, *8*, 611. [CrossRef]
- Ai, G.; Shi, H.; Wu, H.; Li, Z.; Guo, J. The principle of the positioning system based on communication satellites. *Sci. China Ser. G Phys. Mech. Astron.* 2009, 52, 472–488. [CrossRef]

- Madhani, P.H.; Axelrad, P.; Krumvieda, K.; Thomas, J. Application of successive interference cancellation to the GPS pseudolite near-far problem. *IEEE Trans. Aerosp. Electron. Syst.* 2003, 39, 481–488. [CrossRef]
- 96. Wu, T.; Zhan, X.; Zhang, X. Transceiver pseudolite carrier frequency self-alignment closed-loop system. *Aerosp. Syst.* 2020, *3*, 41–52. [CrossRef]
- 97. Hu, Y.; Yu, B.; Song, M.; Deng, Z. Pulse Position Detection of the Pseudo Random Time-Hopping Pseudolite for the Participative GNSS Receivers. *IEEE Access* 2020, *8*, 216151–216161. [CrossRef]
- Liu, X.; Yao, Z.; Lu, M. Robust time-hopping pseudolite signal acquisition method based on dynamic Bayesian network. GPS Solut. 2021, 25, 38. [CrossRef]
- Liu, J.; Zhang, B.; Liu, T.; Xu, G.; Ji, Y.; Sun, M.; Nie, W.; He, Y. An Efficient UD Factorization Implementation of Kalman Filter for RTK Based on Equivalent Principle. *Remote Sens.* 2022, 14, 967. [CrossRef]
- 100. Yun, S.; Yao, Z.; Wang, T.; Lu, M. High accuracy and fast acquisition algorithm for pseudolites-based indoor positioning systems. In Proceedings of the 2016 Fourth International Conference on Ubiquitous Positioning, Indoor Navigation and Location Based Services (UPINLBS), Shanghai, China, 2–4 November 2016; pp. 51–60.
- 101. Klein, D.; Parkinson, B.W. The use of pseudo-satellites for improving GPS performance. J. Inst. Navig. 1984, 31, 303–315. [CrossRef]
- 102. Bai, Z.; Wu, Y.; Wang, W.; Wang, X. A zero intermediate frequency RF transceiver with tunable operating frequency band. *Int. J. RF Microw. Comput.-Aided Eng.* **2021**, *31*, e22534. [CrossRef]
- 103. Borio, D.; Odriscoll, C. Design of a general pseudolite pulsing scheme. IEEE Trans. Aerosp. Electron. Syst. 2014, 50, 2–16. [CrossRef]
- Chen, L.; Kaiqi, L.; Yu, H. Research and analysis on anti-multipath characteristic of pseudolite based on BOC modulation. *Electron.* Des. Eng. 2021, 29, 6. [CrossRef]
- 105. Xu, Y.; Yuan, H.; Wei, D.; Lai, Q.; Zhang, X.; Hao, W. Research on Multi-Source Fusion Based Seamless Indoor/Outdoor Positioning Technology. In Proceedings of the China Satellite Navigation Conference (CSNC) 2015 Proceedings, Xi'an, China, 13–15 May 2015; Volume III, pp. 819–838.
- Maghdid, H.S.; Lami, I.A.; Ghafoor, K.Z.; Lloret, J. Seamless outdoors-indoors localization solutions on smartphones: Implementation and challenges. ACM Comput. Surv. CSUR 2016, 48, 1–34. [CrossRef]
- 107. Liu, K.; Yang, J.; Guo, X.; Zhou, Y.; Liu, C. Correction of fractional cycle bias of pseudolite system for user integer ambiguity resolution. *GPS Solut.* **2018**, 22, 105. [CrossRef]
- Dou, J.; Xu, B.; Dou, L. Impact Assessment of the Asynchronous Clocks Between Reference and User Receivers in Differential Pseudolite Navigation System. *IEEE Sens. J.* 2020, 21, 403–411. [CrossRef]
- 109. Gan, X.; Yu, B.; Huang, L.; Jia, R.; Zhang, H.; Sheng, C.; Fan, G.; Wang, B. Doppler differential positioning technology using the BDS/GPS indoor array pseudolite system. *Sensors* **2019**, *19*, 4580. [CrossRef]
- 110. Sun, Y.; Wang, J.; Chen, J. Indoor precise point positioning with pseudolites using estimated time biases iPPP and iPPP-RTK. *GPS Solut.* **2021**, *25*, 41. [CrossRef]
- He, C.; Yu, B.; Deng, Z. Wireless time synchronization for multiple UAV-borne pseudolites navigation system. In Proceedings of the China Satellite Navigation Conference (CSNC) 2016 Proceedings, Changsha, China, 18–20 May 2016; Volume II, pp. 303–315.
- Bonenberg, L.K.; ROBERTS, G.W.; Matthew, C. Engineering Applications of Integrated Wireless Band Pseudolite and GNSS System. In Proceedings of the FIG Congress, Sydney, NSW, Australia, 11–16 April 2010; pp. 11–16.
- 113. Wang, T.; Yao, Z.; Lu, M. On-the-fly ambiguity resolution involving only carrier phase measurements for stand-alone groundbased positioning systems. *GPS Solut.* **2019**, 23, 36. [CrossRef]
- 114. Sheng, C.; Yu, B.; Zhang, Z.; Fan, G.; Wang, X. An optimal indoor trust-region PPP algorithm with constrain of homologous array pseudolite. *Adv. Space Res.* **2022**, *69*, 1978–1993. [CrossRef]
- 115. Wang, T.; Yao, Z.; Lu, M. Mesh topology based clock synchronization technique for pseudolite systems. *Navigation* **2020**, *67*, 619–632. [CrossRef]
- Guo, X.; Zhou, Y.; Wang, J.; Liu, K.; Liu, C. Precise point positioning for ground-based navigation systems without accurate time synchronization. GPS Solut. 2018, 22, 34. [CrossRef]
- Yu, Y.; Xu, J.; Liang, Y.; Li, F. Long Distance Optical Fiber Time Synchronization Technology and Accuracy Analysis. In Proceedings of the 2021 2nd International Conference on Artificial Intelligence and Education (ICAIE), Dali, China, 18–20 June 2021; pp. 141–149.
- 118. Fu, J.; Li, G.; Tang, T.; Wang, L.; Zhou, Y. Research on Performance Evaluation of Beidou Be Enhanced by Pseudolites. In Proceedings of the China Satellite Navigation Conference, Shanghai, China, 23–25 May 2017; pp. 629–643.
- Liu, H.; Zhao, L.; Huang, X.; Tang, Q. A Technique of Time Synchronization in Pseudolite System Based on Single-difference Method. J. Phys. Conf. Ser. 2021, 1732, 012029. [CrossRef]
- 120. Hwang, S.; Yu, D. Clock synchronization algorithm for pseudolite. Adv. Sci. Technol. Lett. 2013, 44, 36–39.
- 121. Hwang, S.; Yu, D. Clock synchronization of pseudolite using time transfer technique based on GPS code measurement. *Int. J. Softw. Eng. Its Appl.* **2014**, *8*, 35–40.
- Zheng, R. A Pseudo-satellite Implementation Method Using High Precision Time Synchronization. In Proceedings of the China Satellite Navigation Conference (CSNC 2021) Proceedings, Nanchang, China, 26–28 May 2021; pp. 280–288.
- 123. Wang, J.; Lee, H.-K. Impact of pseudolite location errors in positioning. Geomat. Res. Australas. 2002, 77, 81–94.
- 124. Xu, G.; Xu, Y. GPS: Theory, Algorithms and Applications; Springer: Berlin/Heidelberg, Germany, 2016.

- 125. Capderou, M. Handbook of Satellite Orbits: From Kepler to GPS; Springer Science & Business: Berlin/Heidelberg, Germany, 2014.
- 126. Liu, T.; Liu, L.; He, Y.; Sun, M.; Liu, J.; Xu, G. A Theoretical Optimum Tilt Angle Model for Solar Collectors from Keplerian Orbit. *Energies* 2021, 14, 4454. [CrossRef]
- 127. Xu, R.; Chen, W.; Xu, Y.; Ji, S. A new indoor positioning system architecture using GPS signals. *Sensors* **2015**, *15*, 10074–10087. [CrossRef]
- Liu, T.; Xu, T.; Nie, W.; Li, M.; Fang, Z.; Du, Y.; Jiang, Y.; Xu, G. Optimal Independent Baseline Searching for Global GNSS Networks. J. Surv. Eng. 2021, 147, 05020010. [CrossRef]
- 129. Wang, R. A New Ground-Based Pseudolite System Deployment Algorithm Based on MOPSO. Sensors 2021, 21, 5364.
- 130. Zhao, X.; Shuai, Q.; Li, G.; Lu, F.; Zhu, B. A Geometric Layout Method for Synchronous Pseudolite Positioning Systems Based on a New Weighted HDOP. *ISPRS Int. J. Geo-Inf.* **2021**, *10*, 601. [CrossRef]
- 131. Yang, L.; Yang, K.; Sun, D. Research on the Station Layout Method of Ground-Based Pseudolite Positioning System Based on NSGA-II Algorithm. *Wirel. Commun. Mob. Comput.* **2021**, 2021, 1520859. [CrossRef]
- 132. Ma, C.; Yang, J.; Chen, J.; Tang, Y. Indoor and outdoor positioning system based on navigation signal simulator and pseudolites. *Adv. Space Res.* **2018**, *62*, 2509–2517. [CrossRef]
- Wang, X.; Pan, S.; Zhao, Y.; Xia, Y. Propagation Characteristics of Pseudolite Array Signals Indoors. In Proceedings of the China Satellite Navigation Conference, Beijing, China, 22–25 May 2019; pp. 254–265.
- 134. Marathe, T.; Daneshmand, S.; Lachapelle, G. Pseudolite interference mitigation and signal enhancements using an antenna array. In Proceedings of the 2015 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Banff, AB, Canada, 13–16 October 2015; pp. 1–9.
- 135. Alexandru, R.-C.; Elena-Simona, L. Comparison of detection techniques for multipath propagation of pseudolite signals used in dense industrial environments. *Procedia Eng.* 2015, 100, 1294–1300. [CrossRef]
- 136. Gladyshev, A.B.; Dmitriev, D.D.; Veysov, E.; Tyapkin, V.N. A hardware-software complex for modelling and research of near navigation based on pseudolites. *J. Phys. Conf. Ser.* 2017, *803*, 012048. [CrossRef]
- 137. Wang, X.; Pan, S.; Yu, B.; Wang, Y.; Gan, X. Indoor multipath propagation characteristics of BeiDou pseudolite signal. *Bull. Surv. Mapp.* **2019**, *1*, 10.
- 138. Liu, Q.; Huang, Z.; Wang, J. Indoor non-line-of-sight and multipath detection using deep learning approach. *GPS Solut.* **2019**, 23, 75. [CrossRef]
- Fan, C.; Yao, Z.; Xing, J. Portable anechoic chamber-type calibration and the effect of phase corrections on the results of indoor positioning. *GPS Solut.* 2022, 26, 10. [CrossRef]
- Xia, Y.; Pan, S.; Baoguo, Y.U.; Gao, W.; Gan, X.; Wang, X. Asynchronous pseudolite indoor positioning method based on C/N_0 weighted fusion. J. Chin. Inert. Technol. 2019, 27, 154–159.
- 141. Gaochao, Y.; Qing, W.; Baoguo, Y.; Pengfei, L.; Shuang, L. High-precision indoor positioning based on robust LM visual inertial odometer and pseudosatellite. *Acta Geod. Et Cartogr. Sin.* **2022**, *51*, 18.
- 142. Yuanxi, Y. Resilient PNT concept frame. Acta Geod. Et Cartogr. Sin. 2018, 47, 893.
- 143. Li, X.; Zhang, P.; Huang, G.; Zhang, Q.; Guo, J.; Zhao, Y.; Zhao, Q. Performance analysis of indoor pseudolite positioning based on the unscented Kalman filter. *GPS Solut.* **2019**, *23*, 79. [CrossRef]
- 144. Zhao, Y.; Zhang, P.; Guo, J.; Li, X.; Wang, J.; Yang, F.; Wang, X. A new method of high-precision positioning for an indoor pseudolite without using the known point initialization. *Sensors* **2018**, *18*, 1977. [CrossRef]
- 145. Yaming, X.; Fuyu, S.; Peng, Z.; Jinling, W. A Pseudolite Positioning Approach Utilizing Carrier Phase Difference. *Geomat. Inf. Sci. Wuhan Univ.* **2018**, *6*, 1445–1450. [CrossRef]
- 146. Liu, X.; Yao, Z.; Lu, M. A rapid convergent positioning algorithm based on projected cancellation technique for pseudolite positioning systems. *GPS Solut.* 2022, 26, 15. [CrossRef]
- 147. Zhu, R.; Gan, X.; Li, Y.; Zhang, H.; Li, S.; Huang, L. An indoor location method based on optimal DOP of displacement vector components and weighting factor adjustment with multiple array pseudolites. In Proceedings of the 2018 Ubiquitous Positioning, Indoor Navigation and Location-Based Services (UPINLBS), Wuhan, China, 22–23 March 2018; pp. 1–7.
- Yun, S.; Yao, Z.; Wang, T.; Su, J.; Lu, M. On-the-fly ambiguity resolution without requirements of priori position information for passive pseudolite receiver. In Proceedings of the 2019 International Technical Meeting of the Institute of Navigation, Reston, VA, USA, 28–31 January 2019; pp. 598–607.
- 149. Fan, C.; Yao, Z.; Yun, S.; Xing, J. Ground-based PPP-RTK for pseudolite systems. J. Geod. 2021, 95, 133. [CrossRef]
- 150. Yang, Y.; He, H.; Xu, G. Adaptively robust filtering for kinematic geodetic positioning. J. Geod. 2001, 75, 109–116. [CrossRef]
- 151. Zhao, Y.; Guo, J.; Zou, J.; Zhang, P.; Zhang, D.; Li, X.; Huang, G.; Yang, F. A holistic approach to guarantee the reliability of positioning based on carrier phase for indoor pseudolite. *Appl. Sci.* **2020**, *10*, 1199. [CrossRef]
- 152. Li, X.; Huang, G.; Zhang, P.; Zhang, Q. Reliable indoor pseudolite positioning based on a robust estimation and partial ambiguity resolution method. *Sensors* 2019, *19*, 3692. [CrossRef] [PubMed]
- 153. Huang, P.; Rizos, C.; Roberts, C. Airborne Pseudolite Distributed Positioning based on Real-time GNSS PPP. J. Navig. 2019, 72, 1159–1178. [CrossRef]
- 154. Gan, X.; Yu, B.; Wang, X.; Yang, Y.; Jia, R.; Zhang, H.; Sheng, C.; Huang, L.; Wang, B. A new array pseudolites technology for high precision indoor positioning. *IEEE Access* 2019, *7*, 153269–153277. [CrossRef]

- Huang, L.; Gan, X.; Yu, B.; Zhang, H.; Li, S.; Cheng, J.; Liang, X.; Wang, B. An innovative fingerprint location algorithm for indoor positioning based on array pseudolite. *Sensors* 2019, *19*, 4420. [CrossRef]
- 156. Sheng, C.; Gan, X.; Yu, B.; Zhang, J. Precise point positioning algorithm for pseudolite combined with GNSS in a constrained observation environment. *Sensors* **2020**, *20*, 1120. [CrossRef]
- 157. Wang, J.; Dai, L.; Tsujii, T.; Rizos, C.; Grejner-Brzezinska, D.; Toth, C. GPS/INS/Pseudolite integration: Concepts, simulation and testing. In Proceedings of the 14th International Technical Meeting of the Satellite Division of the Institute of Navigation (ION GPS 2001), Salt Lake City, UT, USA, 11–14 September 2001; pp. 2708–2715.
- 158. Lee, H.; Wang, J.; Rizos, C.; Grejner-Brzezinska, D.; Toth, C. GPS/Pseudolite/INS integration: Concept and first tests. *GPS Solut*. **2002**, *6*, 34–46. [CrossRef]
- 159. Grejner-Brzezinska, D.A.; Yi, Y. Experimental GPS/INS/Pseudolite system for kinematic positioning. *Surv. Rev.* 2003, *37*, 113–126. [CrossRef]
- Kim, O.-J.; Hong, D.; Kim, J.; Lee, T.; Kee, C. Experimental Study of Single-Transmitter-Based Precise Indoor Positioning System. IEEE Access 2020, 8, 89919–89934. [CrossRef]
- 161. Grejner-Brzezinska, D.A.; Toth, C.K.; Sun, H.; Wang, X.; Rizos, C. A robust solution to high-accuracy geolocation: Quadruple integration of GPS, IMU, pseudolite, and terrestrial laser scanning. *IEEE Trans. Instrum. Meas.* **2011**, *60*, 3694–3708. [CrossRef]
- Yun, S.; Yao, Z.; Lu, M. On-the-fly ambiguity resolution method for pseudolite/INS integration based on double-difference square observations. GPS Solut. 2021, 25, 137. [CrossRef]
- 163. Lee, H.; Wang, J.; Rizos, C. An integer ambiguity resolution procedure for GPS/pseudolite/INS integration. *J. Geod.* 2005, 79, 242–255. [CrossRef]
- 164. Kim, M.K.; Kim, O.; Kim, Y.S.; Jeon, S.H.; No, H.K.; Shin, B.J.; Kim, J.B.; Kee, C. Pseudolite/Ultra-low-cost IMU Integrated Robust Indoor Navigation System through Real-time Cycle Slip Detection and Compensation. *J. Position. Navig. Timing* **2017**, *6*, 181–194.
- Babu, R.; Wang, J. Ultra-tight GPS/INS/PL integration: A system concept and performance analysis. GPS Solut. 2009, 13, 75–82.
 [CrossRef]
- Gan, X.; Yu, B.; Heng, Z.; Huang, L.; Li, Y. Indoor combination positioning technology of Pseudolites and PDR. In Proceedings of the 2018 Ubiquitous Positioning, Indoor Navigation and Location-Based Services (UPINLBS), Wuhan, China, 22–23 March 2018; pp. 1–7.
- 167. Yan, H.; Yun, Z.; Chen, Y. Indoor LOS Positioning Comparison Between UWB and Pseudo-Lite Technology. *Int. Core J. Eng.* **2021**, 7, 105–113.
- 168. Lee, T.; Kim, C.; Jeon, S.; Jeon, S.; Kim, G.; Kee, C. Pedestrian indoor navigation algorithm based on the pseudolite and low-cost IMU with magnetometers including in-flight calibration. In Proceedings of the 2010 International Technical Meeting of the Institute of Navigation, San Diego, CA, USA, 25–27 January 2010; pp. 230–235.
- Patino-Studencka, L.; Thielecke, J.; Rohmer, G. SLAM for pseudolite system comprising unsynchronised transmitters. In Proceedings of the 2014 Sensor Data Fusion: Trends, Solutions, Applications (SDF), Bonn, Germany, 8–10 October 2014; pp. 1–6.
- 170. Yu, B.; Fan, G.; Luo, Y.; Sheng, C.; Gan, X.; Huang, L.; Rong, Q. Multi-source fusion positioning algorithm based on pseudo-satellite for indoor narrow and long areas. *Adv. Space Res.* 2021, *68*, 4456–4469. [CrossRef]
- 171. Yang, Y. Chinese geodetic coordinate system 2000. Chin. Sci. Bull. 2009, 54, 2714–2721. [CrossRef]
- 172. Lv, Z.; Wei, Z.; Li, J.; Guo, C. Grid model for high-accuracy coordinate transformation of China geodetic coordinate system 2000. J. Geod. Geoinf. Sci. **2020**, 2, 17–25.
- 173. Cheng, P.; Cheng, Y.; Wang, X.; Xu, Y. Update China geodetic coordinate frame considering plate motion. *Satell. Navig.* **2021**, *2*, 2. [CrossRef]
- 174. Yang, Y.; Liu, Y.; Sun, D.; Xu, T.; Xue, S.; Han, Y.; Zeng, A. Seafloor geodetic network establishment and key technologies. *Sci. China Earth Sci.* 2020, *63*, 1188–1198. [CrossRef]
- 175. Zou, Y.; Zhao, W. Making a new area in Xiong'an: Incentives and challenges of China's "Millennium Plan". *Geoforum* **2018**, *88*, 45–48. [CrossRef]
- 176. Zhou, M.; Zhuang, H.; Fang, S. Innovations at Beijing Daxing International: The world's biggest airport terminal. *Proc. Inst. Civ. Eng. Civ. Eng.* **2020**, *173*, 113–118. [CrossRef]
- 177. Chang, C.-L.; Juang, J.-C. Performance analysis of narrowband interference mitigation and near-far resistance scheme for GNSS receivers. *Signal Process.* **2010**, *90*, 2676–2685. [CrossRef]
- 178. Picois, A.V.; Samama, N. Near-far interference mitigation for pseudolites using double transmission. *IEEE Trans. Aerosp. Electron. Syst.* 2014, 50, 2929–2941. [CrossRef]
- 179. Zin, A.; Piccolo, A.; Emmanuele, A.; Siniscalco, L.; Paggi, F.; O'Driscoll, C.; Paonni, M.; Sgammini, M. Mitigating the Near-Far Interference Problem in the GNSS Space Applications. In Proceedings of the 32nd International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GNSS+ 2019), Miami, FL, USA, 16–20 September 2019; pp. 3567–3581.
- Liu, X.; Yao, Z.; Wang, T.; Lu, M. Direct Position Acquisition for Pseudolites Positioning System with Near-far Resistance. In Proceedings of the 2022 International Technical Meeting of The Institute of Navigation, Long Beach, CA, USA, 25–27 January 2022; pp. 957–966.
- Liu, X. Research on Signal Design Method of Pseudolite "Near-Far Effect" Based on TDMA Technique. In Proceedings of the China Satellite Navigation Conference, Shanghai, China, 23–25 May 2017; pp. 417–429.

- 182. Börner, A.; Baumbach, D.; Buder, M.; Choinowski, A.; Ernst, I.; Funk, E.; Grießbach, D.; Schischmanow, A.; Wohlfeil, J.; Zuev, S. IPS—A vision aided navigation system. *Adv. Opt. Technol.* 2017, *6*, 121–129. [CrossRef]
- 183. Dai, L.; Wang, J.; Tsujii, T.; Rizos, C. Pseudolite applications in positioning and navigation: Modelling and geometric analysis. In Proceedings of the International Symposium on Kinematic Systems in Geodesy, Geomatics & Navigation (KIS2001), Banff, AB, Canada, 5–8 June 2001; pp. 482–489.
- Wang, H.-H.; Zhai, C.-R.; Zhan, X.-Q.; He, Z. Outdoor navigation system using integrated gps and pseudolite signals: Theoretical analysis and simulation. In Proceedings of the 2008 International Conference on Information and Automation, Changsha, China, 20–23 June 2008; pp. 1127–1131.
- 185. Yuanxi, Y.; Cheng, Y.; Xia, R. PNT intelligent services. Acta Geod. Et Cartogr. Sin. 2021, 50, 1006.
- 186. Liu, J.; Luo, Y.; Guo, C.; Gao, K. PNT intelligence and intelligent PNT. Acta Geod. Et Cartogr. Sin. 2022, 51, 811.
- 187. Li, J.; Chen, W.; Yao, J.; Wang, Y. Research and Application of PNT Information Service System Based on "End+ Network+ Cloud". In Proceedings of the China Satellite Navigation Conference, Shanghai, China, 23–25 May 2017; pp. 366–375.
- Petrovski, I.; Okano, K.; Ishii, M.; Torimoto, H.; Konishi, Y.; Shibasaki, R. Pseudolite implementation for social infrastructure and seamless Indoor/Outdoor positioning. In Proceedings of the 15th International Technical Meeting of the Satellite Division of The Institute of Navigation (ION GPS 2002), Portland, OR, USA, 24–27 September 2002; pp. 69–76.
- Qiu, K.; Chen, R.; Huang, H. A Practical Indoor and Outdoor Seamless Navigation System Based on Electronic Map and Geomagnetism. In Proceedings of the 2021 13th International Conference on Machine Learning and Computing, New York, NY, USA, 26 February–1 March 2021; pp. 588–594.
- Korb, M.; Huang, Q.; Stockel, P.; Kappen, G.C.; Weber, B.; Garcia, M. A Cellular-Modem-Hosted Low-Cost Single-Shot Dual-Mode Assisted-GNSS Receiver for the Internet of Things. In Proceedings of the 2020 IEEE/ION Position, Location and Navigation Symposium (PLANS), Portland, OR, USA, 20–23 April 2020; pp. 1273–1279.
- 191. Schmitt, C.; Meier, J.; Diez, M.; Stiller, B. OTIoT—A browser-based object tracking solution for the Internet of Things. In Proceedings of the 2018 IEEE 4th World Forum on Internet of Things (WF-IoT), Singapore, 5–8 February 2018; pp. 445–451.
- 192. Guillemin, P.; Berens, F.; Carugi, M.; Arndt, M.; Ladid, L.; Percivall, G.; De Lathouwer, B.; Liang, S.; Bröring, A.; Thubert, P. Internet of Things Standardisation—Status, Requirements, Initiatives and Organisations; River Publishers: Aalborg, Denmark, 2013; p. 259.
- 193. Kassas, Z.M.; Closas, P.; Gross, J. Navigation systems panel report navigation systems for autonomous and semi-autonomous vehicles: Current trends and future challenges. *IEEE Aerosp. Electron. Syst. Mag.* **2019**, *34*, 82–84. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.